\documentclass[11pt]{article}

\usepackage{amsmath, amsthm, amsfonts,color}
\usepackage[margin=1in]{geometry}
%\usepackage{fullpage}

% Theorems
%-----------------------------------------------------------------
\newtheorem{thm}{Theorem}[section]
\newtheorem{cor}[thm]{Corollary}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{prop}[thm]{Proposition}
\theoremstyle{definition}
\newtheorem{defn}[thm]{Definition}
\theoremstyle{remark}
\newtheorem{rem}[thm]{Remark}

\def\RR{\mathbb{R}}
\def\ZZ{\mathbb{Z}}
\def\qed{\hfill $\Box$\par\vskip3mm}

%\newcommand{\abs}[1]{\left\vert#1\right\vert}

\title{Combined concave-convex effects in anisotropic elliptic equations with variable exponent}
\author{Vicen\c tiu D. R\u adulescu$\,^1$ and Ionela-Loredana St\u{a}ncu\c{t}$\,^2$\\
  \footnotesize $^1\,$ Department of Mathematics, Faculty of Science, King Abdulaziz University,
Jeddah, Saudi Arabia\\
  \footnotesize $^2\,$Department of Mathematics, University of Craiova, 200585 Craiova, Romania}

\begin{document}
\begin{thebibliography}{99}
{\footnotesize
\bibitem{bib10} S. Alama and G. Tarantello, \emph{Elliptic problems with nonlinearities indefinite in sign}, J. Funct. Anal. {\bf 141} (1996), 159-215.

    \bibitem{abcpaper} A. Ambrosetti, H. Brezis, and G. Cerami, \emph{ Combined effects of concave and convex nonlinearities in some elliptic problems}, J. Funct. Anal. {\bf 122} (1994), 519-543.

\bibitem{bib8}  A. Ambrosetti and P.H. Rabinowitz, \emph{Dual variational methods in critical point theory}, J. Funct. Anal. {\bf 14} (1973), 349-381.

\bibitem{aupucci1}  G. Autuori and P. Pucci, \emph{Existence of entire solutions for a class of quasilinear elliptic equations}, NoDEA Nonlinear Differential Equations Appl. {\bf 20} (2013), 977-1009.

\bibitem{aupucci2}  G. Autuori and P. Pucci, \emph{Elliptic problems involving the fractional Laplacian in ${\mathbb R}^N$}, J. Differential Equations {\bf 255} (2013), 2340-2362.

\bibitem{brandle} C. Br\"andle, E. Colorado, A. de Pablo, and U. S\'anchez,
\emph{A concave-convex elliptic problem involving the fractional Laplacian},
Proc. Roy. Soc. Edinburgh Sect. A {\bf 143} (2013), 39-71.

\bibitem{bib17} H. Brezis, \emph{Analyse fonctionnelle: th\'{e}orie et applications}, Masson, Paris, 1992.

\bibitem{bib11} J. Chabrowski, \emph{Elliptic variational problems with indefinite nonlinearities}, Topological Methods in Nonlinear Analysis {\bf 9} (1997), 221-231.

\bibitem{bib7}  D. Gilbarg and N. S. Trudinger, \emph{Elliptic Partial Differential Equations of Second Order},
Springer, Berlin, 1998.

\bibitem{bib1}  P. Harjulehto, \emph{Variable exponent Sobolev spaces with zero boundary values}, Math. Bohem. {\bf 132} (2007), 125-136.

\bibitem{bib2}  P. Harjulehto, P. H\"{a}st\"{o}, M. Koskenoja, and S. Varonen, \emph{The Dirichlet energy integral and variable exponent Sobolev spaces with zero boundary values}, Potential Anal. {\bf 25} (2006), 79-94.

\bibitem{bib3}  P. H\"{a}st\"{o}, \emph{On the density of continuous functions in variable exponent Sobolev spaces}, Rev. Mat. Iberoamericana {\bf 23} (2007), 74-82.

\bibitem{bib4}  O. Kov\'{a}\v{c}ik, J. R\'{a}kosn\'{\i}k, \emph{On spaces $L^{p(x)}$ and $W^{k,p(x)}$}, Czechoslovak Math. J. {\bf 41} (1991), 592-618.

\bibitem{bib15} P. Pucci and V. R\u{a}dulescu, \emph{Combined effects in quasilinear elliptic problems
    with lack of compactness}, Rend. Lincei Mat. Appl. {\bf 22} (2011), 189-205.

\bibitem{bib14} V. R\u{a}dulescu,  \emph{Combined effects for a stationary problem with indefinite nonliniarities and lack of compactness}, Dynamic Systems and Applications {\bf 22} (2013), 371-384.

\bibitem{bib19} J. Simon, \emph{R\'{e}gularit\'{e} de la solution d'une \'{e}quation non lin\'{e}aire dans $\RR^{N}$}, in \emph{Journ\'{e}es d'Analyse Non Lin\'{e}aire}, Ph. B\'{e}nilan and J. Robert, eds., Vol. 665, Lecture Notes in Mathematics, Springer, Berlin, 1978, pp. 205-227.

\bibitem{bib6}  M. Struwe, \emph{Variational Methods: Applications to Nonlinear Partial Differential Equations and Hamiltonian Systems}, Springer, Heidelberg, 1996.

\bibitem{bib12} S. B. Tshinanga, \emph{On multiple solutions of semilinear elliptic equations on unbounded domains with concave and convex nonlinearities}, Nonlinear Analysis: Theory, Methods and Applications
    {\bf 28} (1997), 809-814.

\bibitem{bib9}  M. Willem, \emph{Minimax Theorems}, Birkh\"{a}user, Boston, 1996.

\bibitem{bib13} J. Yang and X. Zhu, \emph{On the existence of nontrivial solutions of a quasilinear elliptic boundary value problem for unbounded domains. (I) Positive mass case}, Acta Math. Sci. {\bf 7} (1987), 341-359.
}

\end{thebibliography}

\maketitle

\begin{abstract}
We establish the existence and multiplicity of solutions for a class of quasilinear elliptic equations involving the anisotropic $\vec{p}(\cdot)$-Laplace operator, on a bounded domain with smooth boundary. We work on the weighted anisotropic variable exponent Sobolev space and our main tools are Sobolev embeddings and the mountain pass theorem.\\
\textbf{Keywords:} quasilinear elliptic equations; $\vec{p}(\cdot)$-Laplace operator; weighted anisotropic variable exponent Sobolev space; critical point; weak solution.\\
{\bf 2010 Mathematics Subject Classification}: 35J60; 58E05.
\end{abstract}


\section{Introduction}

In this paper, we are interested in the existence and multiplicity of weak solutions for the nonhomogeneous anisotropic eigenvalue problem
\label{pb}
\begin{equation}
\left \{
\begin{array}{l l}
-\displaystyle\sum_{i=1}^{N}\partial_{x_{i}}\big(|\partial_{x_{i}}u|^{p_{i}(x)-2}\partial_{x_{i}}u\big)
=\lambda|u|^{q(x)-2}u-h(x)|u|^{r(x)-2}u\ & \text{in}\ \Omega,\\
u=0\ & \text{on}\ \partial\Omega,\\
\end{array} \right. \label{l1}
\end{equation}
where $\Omega\subset\RR^{N}$ ($N\geq3$) is a bounded domain with smooth boundary, $\lambda>0$ is a real number, $p_{i},q,r$ are continuous functions on $\overline{\Omega}$ such that $2\leq p_{i}(x)<N$, $2<q(x)<r(x)$ for any $x\in\overline{\Omega}$ and $i\in\{1,...,N\}$, and $h:\overline{\Omega}\rightarrow \RR$ is a continuous positive function satisfying the conditions
\begin{equation}
\label{h1}
\int_{\Omega}\lambda^{\frac{r(x)}{r(x)-q(x)}}\frac{1}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}dx<\infty
\end{equation}
and
\begin{equation}
\label{h2}
\int_{\Omega}\Bigg(\frac{\lambda}{h(x)^{\frac{q(x)-2}{r(x)-2}}}\Bigg)^{\frac{r(x)}{r(x)-q(x)}}dx<\infty.
\end{equation}

We first establish the non-existence of nontrivial weak solutions for problem \eqref{pb} if $\lambda$ is small enough and then we get the existence of at least two nontrivial weak solutions for problem \eqref{pb} if $\lambda$ is sufficiently large.

We start with preliminary basic notations and terminology on the theory of \textit{Lebesgue--Sobolev
spaces with variable exponent}.\\
Set $$C_{+}(\overline{\Omega})=\{g; \ g\in C(\overline{\Omega}), \ g(x)>1 \text{ for all } x\in\overline{\Omega}\}.$$
For any $g\in C_{+}(\overline{\Omega})$ we define
$$g^{+}=\sup_{x\in\Omega}g(x)\ \text{ and } \ g^{-}=\inf_{x\in\Omega}g(x).$$
For any $p\in C_{+}(\overline{\Omega})$ we define the \textit{variable exponent Lebesgue space}
\begin{equation*}
L^{p(\cdot)}(\Omega)=\bigg\{u;\ u \text{ is a measurable real-valued function and } \int_{\Omega}|u|^{p(x)}dx<\infty\bigg\},
\end{equation*}
endowed with the so-called \textit{Luxemburg norm}
$$|u|_{p(\cdot)}=\inf\bigg\{\mu>0; \ \int_{\Omega}\bigg|\frac{u(x)}{\mu}\bigg|^{p(x)}dx\leq1\bigg\},$$
which is a separable and reflexive Banach space. If $0<|\Omega|<\infty$ and $p_{1},p_{2}$ are variable exponents in $C_{+}(\overline{\Omega})$ such that $p_{1}(x)\leq p_{2}(x)$ almost everywhere in $\Omega$, then the embedding $L^{p_{2}(\cdot)}(\Omega)\hookrightarrow L^{p_{1}(\cdot)}(\Omega)$ is continuous.

We denote by $L^{p'(\cdot)}(\Omega)$ the conjugate space of $L^{p(\cdot)}(\Omega)$, where $\frac{1}{p(x)}+\frac{1}{p'(x)}=1$. For any $u\in L^{p(\cdot)}(\Omega)$ and $v\in L^{p'(\cdot)}(\Omega)$ the following H\"{o}lder type inequality
\begin{equation}
\label{Holder}
\bigg|\int_{\Omega}uv \,dx\bigg|\leq\bigg(\frac{1}{p^{-}}+\frac{1}{p'^{-}}\bigg)|u|_{p(\cdot)}|v|_{p'(\cdot)}
\end{equation}
holds true.

An important role in handling the generalized Lebesgue spaces is played by the $p(\cdot)$-\textit{modular} of the $L^{p(\cdot)}(\Omega)$ space, which is the mapping $\rho_{p(\cdot)}:L^{p(\cdot)}(\Omega)\rightarrow\RR$ defined by
$$\rho_{p(\cdot)}(u)=\int_{\Omega}|u|^{p(x)}dx.$$
If $(u_{n}), u\in L^{p(\cdot)}(\Omega)$, then the following relations hold true:
\begin{equation}
|u|_{p(\cdot)}<1 \ (=1; \ >1)\Leftrightarrow \rho_{p(\cdot)}(u)<1 \ (=1; \ >1),
\end{equation}
\begin{equation}
\label{|u|>1}
|u|_{p(\cdot)}>1 \Rightarrow |u|_{p(\cdot)}^{p^{-}}\leq\rho_{p(\cdot)}(u)\leq|u|_{p(\cdot)}^{p^{+}},
\end{equation}
\begin{equation}
\label{|u|<1}
|u|_{p(\cdot)}<1 \Rightarrow |u|_{p(\cdot)}^{p^{+}}\leq\rho_{p(\cdot)}(u)\leq|u|_{p(\cdot)}^{p^{-}},
\end{equation}
\begin{equation}
\label{conv 0}
|u_{n}-u|_{p(\cdot)}\rightarrow0\Leftrightarrow \rho_{p(\cdot)}(u_{n}-u)\rightarrow0.
\end{equation}

For proofs and  further related properties of variable exponent Lebesgue spaces we refer to \cite{bib4}.

By $L^{r(\cdot)}_{h}(\Omega)$, where $r:\overline{\Omega}\rightarrow(1,\infty)$ (with $r^{+}<+\infty$) and $h:\overline{\Omega}\rightarrow[0,\infty)$ are continuous functions, we denote the weighted Lebesgue space
$$L^{r(\cdot)}_{h}(\Omega)=\bigg\{u;\ u \text{ is a measurable real-valued function and } \int_{\Omega}h(x)|u|^{r(x)}dx<\infty\bigg\},$$
equipped with the norm
$$|u|_{h,r(\cdot)}=\inf\bigg\{\mu>0; \ \int_{\Omega}h(x)\bigg|\frac{u(x)}{\mu}\bigg|^{r(x)}dx\leq1\bigg\}.$$
We notice that, if $h(x)\equiv1$ on $\RR^{N}$, the resulting norm is even $|\cdot|_{r(\cdot)}$.

We denote by $W_{0}^{1,p(\cdot)}(\Omega)$ the \textit{variable exponent Sobolev space} defined by
$$W^{1,p(\cdot)}_{0}(\Omega)=\big\{u; \ u_{\mid\partial\Omega}=0, \ u\in L^{p(\cdot)}(\Omega)\text{ and }|\nabla u|\in L^{p(\cdot)}(\Omega)\big\},$$
endowed with the equivalent norms
$$\|u\|_{p(\cdot)}=|u|_{p(\cdot)}+|\nabla u|_{p(\cdot)}$$
and
$$\|u\|=\inf\bigg\{\mu>0; \ \int_{\Omega}\bigg(\bigg|\frac{\nabla u(x)}{\mu}\bigg|^{p(x)}+\bigg|\frac{u(x)}{\mu}\bigg|^{p(x)}\bigg)dx\leq1\bigg\},$$
where, in the definition of $\|u\|_{p(\cdot)}$, $|\nabla u|_{p(\cdot)}$ is the Luxemburg norm of $|\nabla u|$. We remember that $W_{0}^{1,p(\cdot)}(\Omega)$ is a separable and reflexive Banach space. Also, we note that if $s\in C_{+}(\overline{\Omega})$ and $s(x)<p^{*}(x)$ for all $x\in\overline{\Omega}$, then the embedding $W_{0}^{1,p(\cdot)}(\Omega)\hookrightarrow L^{s(\cdot)}(\Omega)$ is compact and continuous, where $p^{*}(x)=\frac{Np(x)}{N-p(x)}$ and $p(x)<N$.

We refer to \cite{bib1,bib2,bib3,bib4} for more properties, details, extensions and further
references.

Finally, we present the \emph{anisotropic variable exponent Sobolev space} $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$, where $\vec{p}:\overline{\Omega}\rightarrow\RR^{N}$ is the vectorial function
$\vec{p}(\cdot)=(p_{1}(\cdot),...,p_{N}(\cdot))$
and the components $p_{i}\in C_{+}(\overline{\Omega})$, $i\in\{1,...,N\}$, are logarithmic H\"{o}lder continuous, that is, there exists $M>0$ such that $|p_{i}(x)-p_{i}(y)|\leq-M/\text{log}(|x-y|)$ for any $x,y\in\Omega$ with $|x-y|\leq1/2$ and $i\in\{1,...,N\}$. $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ is the closure of $C_{0}^{\infty}(\Omega)$ under the norm
$$\|u\|_{\vec{p}(\cdot)}=\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}$$
and is a reflexive Banach space. This space is a natural generalization of the variable exponent
Sobolev space $W_{0}^{1,p(\cdot)}(\Omega).$

Now, we introduce $\vec{P}_{+}, \vec{P}_{-}\in\RR^{N}$ as
$$\vec{P}_{+}=(p_{1}^{+},...,p_{N}^{+}), \hspace{0.5cm} \vec{P}_{-}=(p_{1}^{-},...,p_{N}^{-}),$$
and $P_{+}^{+}, P_{-}^{+}, P_{-}^{-}\in\RR^{+}$ as
$$P_{+}^{+}=\max\{p_{1}^{+},...,p_{N}^{+}\}, \hspace{0.5cm}  P_{-}^{+}=\max\{p_{1}^{-},...,p_{N}^{-}\}, \hspace{0.5cm} P_{-}^{-}=\min\{p_{1}^{-},...,p_{N}^{-}\}.$$
We also always assume that
$$\sum_{i=1}^{N}\frac{1}{p_{i}^{-}}>1,$$
and define $P_{-}^{*}, \ P_{-,\infty}\in\RR^{+}$ by
$$P_{-}^{*}=\frac{N}{\sum_{i=1}^{N}1/p_{i}^{-}-1}, \hspace{0.5cm} P_{-,\infty}=\max\{P_{-}^{+}, P_{-}^{*}\}.$$


\section{Main results}

In this paper we seek weak solutions for problem \eqref{pb} in a subspace of $W^{1,\vec{p}(\cdot)}_{0}(\Omega)$. Let $E$ be the \textit{weighted anisotropic variable exponent Sobolev space} defined by
$$E=\bigg\{u\in W^{1,\vec{p}(\cdot)}_{0}(\Omega); \ \int_{\Omega}h(x)|u|^{r(x)}dx<\infty\bigg\},$$
equipped with the norm
$$\|u\|_{E}=\|u\|_{\vec{p}(\cdot)}+|u|_{h,r(\cdot)}.$$

By a \textit{weak solution} for problem \eqref{pb} we understand a function $u\in E$ with $u(x)=0$ almost everywhere on $\partial\Omega$ so that
\begin{equation}
\label{sol}
\int_{\Omega}\Bigg\{\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)-2}\partial_{x_{i}}u\partial_{x_{i}}v -\lambda|u|^{q(x)-2}uv+h(x)|u|^{r(x)-2}uv\Bigg\}dx=0
\end{equation}
for all $u,v\in E$.

Let $\Phi:E\rightarrow\RR$ be the energy functional defined by
$$\Phi(u)=\int_{\Omega}\Bigg\{\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}-
\frac{\lambda}{q(x)}|u|^{q(x)}+\frac{h(x)}{r(x)}|u|^{r(x)}\Bigg\}dx.$$
By standard arguments we have $\Phi\in C^{1}(E,\RR)$ and the derivative is given by
$$\langle\Phi'(u),v\rangle
=\int_{\Omega}\Bigg\{\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)-2}\partial_{x_{i}}u\partial_{x_{i}}v
-\lambda|u|^{q(x)-2}uv+h(x)|u|^{r(x)-2}uv\Bigg\}dx$$
for all $u,v\in E$. Thus, the weak solutions of problem \eqref{pb} coincide with the critical points of $\Phi$.

\begin{thm}
\label{thm1}
Assume that the function $q\in C(\overline{\Omega})$ verifies the hypothesis
$$\max_{x\in\overline{\Omega}}q(x)<P_{-,\infty}.$$
Then there exists $\lambda^{*}>0$ such that for any $\lambda\in(0,\lambda^{*}]$ problem \eqref{pb} does not have a nontrivial weak solution.
\end{thm}

\begin{thm}
\label{thm2}
Assume that $q\in C(\overline{\Omega})$ and $r\in C(\overline{\Omega})$ satisfy the inequalities\\
$$\max_{x\in\overline{\Omega}}q(x)<P_{-,\infty}, \hspace{0.5cm} \max_{x\in\overline{\Omega}}r(x)<P_{-,\infty} \hspace{0.5cm} \text{and} \hspace{0.5cm} P_{+}^{+}<q^{-}.$$
Then there exists $\lambda_{0}>0$ such that for $\lambda>\lambda_{0}$ problem \eqref{pb} possesses at least two nontrivial weak solutions.
\end{thm}

\begin{rem}
Similar results as those presented above were obtained in the linear case, for Laplace equations, in \cite{bib10}, \cite{bib11}, \cite{bib12}. Also, for $p$-Laplace equations, for a general case these can be found in  \cite{bib13}, and for a particular case we refer to \cite{bib15} and \cite{bib14}.
\end{rem}

The arguments in our proofs are inspired by the methods developed in the paper by Pucci and R\u adulescu \cite{bib15}. However, in the present paper there are several technical difficulties that are mainly due to the presence of several variable exponents. For instance, the standard energy estimates follow by more delicate computation as well as the standard $L^p$-estimates; in our case, due to variable exponents, it is often more efficient to use the $p(\,\cdot\,)$-modular instead of the standard $L^p$ norm.

We conclude this section by mentioning the pioneering paper by Ambrosetti, Brezis and Cerami \cite{abcpaper} on concave-convex nonlinearities in the semilinear case. This result was extended by Autuori and Pucci \cite{aupucci1} to the quasilinear case and by Autuori and Pucci \cite{aupucci2} and Br\"andle,  Colorado,  de Pablo, and  S\'anchez \cite{brandle} to elliptic equations involving the fractional Laplacian.

\section{Proof of Theorem \ref{thm1}}

We assume by contradiction that $u\in E$ is a weak solution of problem \eqref{pb}. Taking $v=u$ in \eqref{sol} we get
\begin{equation}
\label{u=v}
\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx+\int_{\Omega}h(x)|u|^{r(x)}dx
=\lambda\int_{\Omega}|u|^{q(x)}dx.
\end{equation}
Let us recall now the Young's inequality
$$ab\leq\frac{a^{\alpha}}{\alpha}+\frac{b^{\beta}}{\beta}, \ \ \forall a,b>0$$
where $\alpha,\beta>1$ fulfill $\frac{1}{\alpha}+\frac{1}{\beta}=1$.

Put $a=h(x)^{\frac{q(x)}{r(x)}}|u|^{q(x)}$, $b=\frac{\lambda}{h(x)^{\frac{q(x)}{r(x)}}}$, $\alpha=\frac{r(x)}{q(x)}$ and $\beta=\frac{r(x)}{r(x)-q(x)}$ so that
$$\lambda|u|^{q(x)}\leq\frac{q(x)}{r(x)}h(x)|u|^{r(x)}+
\frac{r(x)-q(x)}{r(x)}\cdot\frac{\lambda^{\frac{r(x)}{r(x)-q(x)}}}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}.$$
Taking into account that $\frac{q(x)}{r(x)}<1$ and $\frac{r(x)-q(x)}{r(x)}<1$ and integrating over $\Omega$ we obtain
$$\lambda\int_{\Omega}|u|^{q(x)}dx<\int_{\Omega}h(x)|u|^{r(x)}dx+
\int_{\Omega}\lambda^{\frac{r(x)}{r(x)-q(x)}}\frac{1}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}dx.$$
Considering the last inequality and \eqref{u=v} we get
\begin{equation}
\label{sum-lambda-h}
\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx<\int_{\Omega}\lambda^{\frac{r(x)}{r(x)-q(x)}}
\frac{1}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}dx.
\end{equation}
Clearly $\displaystyle\int_{\Omega}h(x)|u|^{r(x)}dx\geq0$, which together with \eqref{u=v} yield that
\begin{equation}
\label{sum-lambda}
\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx\leq\lambda\int_{\Omega}|u|^{q(x)}dx.
\end{equation}
On the other hand,  the embedding
$W_{0}^{1,\vec{p}(\cdot)}(\Omega)\hookrightarrow L^{q(\cdot)}(\Omega)$ is continuous. Thus, there exists a constant $C>1$ such that
\begin{equation}
\label{|u|<C||u||}
|u|_{q(\cdot)}\leq C\|u\|_{\vec{p}(\cdot)}, \ \forall u\in W_{0}^{1,\vec{p}(\cdot)}(\Omega).
\end{equation}
Let $u\in W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ be such that $\|u\|_{\vec{p}(\cdot)}<\frac{1}{C}$.
Then \eqref{|u|<C||u||} implies
$$|u|_{q(\cdot)}<1 \ \text{for all} \ u\in W_{0}^{1,\vec{p}(\cdot)}(\Omega) \ \text{with} \ \|u\|_{\vec{p}(\cdot)}<\frac{1}{C}.$$
Hence, by property \eqref{|u|<1} we have
\begin{equation}
\label{int |u|<}
\int_{\Omega}|u|^{q(x)}dx\leq|u|_{q(\cdot)}^{q^{-}}.
\end{equation}
Combining \eqref{sum-lambda}, \eqref{|u|<C||u||} and \eqref{int |u|<} it follows that
\begin{equation}
\label{int sum <lambda...}
\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx\leq\lambda C^{q^{-}}\|u\|_{\vec{p}(\cdot)}^{q^{-}}.
\end{equation}
Also, $\|u\|_{\vec{p}(\cdot)}<1$ yields $|\partial_{x_{i}}u|_{p_{i}(\cdot)}<1$ and thus,
by \eqref{|u|<1}, we infer that
\begin{equation}
\label{sum p_i+<int p_i(x)}
\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}^{p_{i}^{+}}
\leq\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx.
\end{equation}
If we apply the Jensen inequality to the convex function $a:\RR^{+}\rightarrow\RR^{+}$, $a(t)=t^{P_{+}^{+}}$,
$P_{+}^{+}\geq2$, we get
\begin{equation}
\label{Jensen}
\frac{\|u\|_{\vec{p}(\cdot)}^{P_{+}^{+}}}{N^{P_{+}^{+}-1}}
=N\Bigg(\frac{\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}}{N}\Bigg)^{P_{+}^{+}}
\leq\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}^{P_{+}^{+}}
\leq\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}^{p_{i}^{+}}.
\end{equation}
Taking in consideration \eqref{sum p_i+<int p_i(x)} and \eqref{Jensen} we deduce that
\begin{equation}
\label{techniques}
\|u\|_{\vec{p}(\cdot)}^{P_{+}^{+}}
\leq N^{P_{+}^{+}-1}\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx.
\end{equation}
By a simple computation, using \eqref{int sum <lambda...} and \eqref{techniques}, one has
\begin{equation*}
\frac{1}{\Bigg(\lambda C^{q^{-}}N^{\frac{q^{-}(P_{+}^{+}-1)}{P_{+}^{+}}}\Bigg)^{\frac{P_{+}^{+}}{q^{-}-P_{+}^{+}}}}
\leq\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx.
\end{equation*}
This last inequality together with \eqref{sum-lambda-h} lead us to
\begin{equation*}
\lambda>\bigg(CN^{\frac{P_{+}^{+}-1}{P_{+}^{+}}}\bigg)^{-q^{-}}
\Bigg(\int_{\Omega}\lambda^{\frac{r(x)}{r(x)-q(x)}}\frac{1}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}dx\Bigg)^
{\frac{P_{+}^{+}-q^{-}}{P_{+}^{+}}}
\end{equation*}
for any $u\in E$ with $\|u\|_{\vec{p}(\cdot)}<\frac{1}{C}$. Denoting the term in the right-hand side of the previous
inequality by $\lambda^{*}$ , we conclude that Theorem \ref{thm1} holds true.\qed

\section{Proof of Theorem \ref{thm2}}

First, we will establish some auxiliary results.
\begin{lem}
\label{lemma1}
The energy functional $\Phi$ is coercive on $E$.
\end{lem}

{\it Proof.}
The proof of this Lemma relies on the inequality:

For any $k_{1},k_{2}>0$ and $0<q<r$, a straightforward computation shows that
\begin{equation}
\label{k1k2}
k_{1}|t|^{q}-k_{2}|t|^{r}\leq C k_{1}\bigg(\frac{k_{1}}{k_{2}}\bigg)^{\frac{q}{r-q}}, \ \ \forall t\in\RR,
\end{equation}
where $C>0$ is a constant depending on $q$ and $r$.

If we choose in \eqref{k1k2} $q=q(x)$, $r=r(x)$, $k_{1}=\frac{\lambda}{q(x)}$ and $k_{2}=\frac{h(x)}{2r(x)}$ we obtain
\begin{align*}
\frac{\lambda}{q(x)}|u|^{q(x)}-\frac{h(x)}{2r(x)}|u|^{r(x)}&\leq C\frac{\lambda}{q(x)}\bigg(\frac{\lambda}{q(x)}\cdot\frac{2r(x)}{h(x)}\bigg)^{\frac{q(x)}{r(x)-q(x)}}\\
&=C\bigg(\frac{1}{q(x)}\bigg)^{\frac{r(x)}{r(x)-q(x)}}\big(2r(x)\big)^{^{\frac{q(x)}{r(x)-q(x)}}}
\lambda^{^{\frac{r(x)}{r(x)-q(x)}}}\frac{1}{h(x)^\frac{q(x)}{r(x)-q(x)}}.
\end{align*}
But the terms $\big(\frac{1}{q(x)}\big)^{\frac{r(x)}{r(x)-q(x)}}$, $(2r(x))^{\frac{q(x)}{r(x)-q(x)}}$ are bounded
(since $2<q(x)<r(x)<P_{-,\infty}$), so the above inequality becomes
$$\frac{\lambda}{q(x)}|u|^{q(x)}-\frac{h(x)}{2r(x)}|u|^{r(x)}\leq C_{1}\lambda^{^{\frac{r(x)}{r(x)-q(x)}}}\frac{1}{h(x)^\frac{q(x)}{r(x)-q(x)}}.$$
Integrating the previous inequality over $\Omega$ and taking into account the hypothesis \eqref{h1} we deduce that there exists a constant $C_{2}>0$ so that
$$\int_{\Omega}\bigg(\frac{\lambda}{q(x)}|u|^{q(x)}-\frac{h(x)}{2r(x)}|u|^{r(x)}\bigg)dx\leq C_{2}.$$
Thus
\begin{align}
\label{Phi is bounded}
\Phi(u)&=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx
-\lambda\int_{\Omega}\frac{1}{q(x)}|u|^{q(x)}dx+\int_{\Omega}\frac{h(x)}{r(x)}|u|^{r(x)}dx\nonumber\\
&=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx
-\int_{\Omega}\bigg(\frac{\lambda}{q(x)}|u|^{q(x)}-\frac{h(x)}{2r(x)}|u|^{r(x)}\bigg)dx\nonumber\\
&\quad{}-\frac{1}{2}\int_{\Omega}\frac{h(x)}{r(x)}|u|^{r(x)}dx+\int_{\Omega}\frac{h(x)}{r(x)}|u|^{r(x)}dx\nonumber\\
&\geq\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx
+\frac{1}{2}\int_{\Omega}\frac{h(x)}{r(x)}|u|^{r(x)}dx-C_{2}\nonumber\\
&\geq\frac{1}{P_{+}^{+}}\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx
+\frac{1}{2r^{+}}\int_{\Omega}h(x)|u|^{r(x)}dx-C_{2}.
\end{align}
To proceed further, let $v\in L^{r(\cdot)}(\Omega)$, $r\in C_{+}(\overline{\Omega})$,
provided that $|v|_{r(\cdot)}>1$. By relation \eqref{|u|>1} we get
$$|v|_{r(\cdot)}^{r^{-}}\leq\int_{\Omega}|v|^{r(x)}dx\leq|v|_{r(\cdot)}^{r^{+}}.$$
Taking $v(x)=h(x)^{\frac{1}{r(x)}}u(x)$ we see that
\begin{equation}
\label{|u|_{h,r(x)}}
|u|_{h,r(\cdot)}^{r^{-}}\leq\int_{\Omega}h(x)|u|^{r(x)}dx\leq|u|_{h,r(\cdot)}^{r^{+}}.
\end{equation}
Also, let $u\in E$ be such that $|\partial_{x_{i}}u|_{p_{i}(\cdot)}>1$ for all $i\in\{1,...,N\}$, whence
$\|u\|_{\vec{p}(\cdot)}>1$. By \eqref{|u|>1} we have
\begin{equation}
\label{sum p_i-<int p_i(x)}
\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}^{p_{i}^{-}}
\leq\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx.
\end{equation}
Applying Jensen's inequality to the convex function $b:\RR^{+}\rightarrow\RR^{+}$, $b(t)=t^{P_{-}^{-}}$, $P_{-}^{-}\geq2$,
we find that
\begin{equation}
\label{Jensen'}
\frac{\|u\|_{\vec{p}(\cdot)}^{P_{-}^{-}}}{N^{P_{-}^{-}-1}}
=N\Bigg(\frac{\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}}{N}\Bigg)^{P_{-}^{-}}
\leq\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}^{P_{-}^{-}}
\leq\sum_{i=1}^{N}|\partial_{x_{i}}u|_{p_{i}(\cdot)}^{p_{i}^{-}}.
\end{equation}
Consequently, \eqref{Phi is bounded}, \eqref{|u|_{h,r(x)}}, \eqref{sum p_i-<int p_i(x)} and \eqref{Jensen'} yield to
\begin{align*}
\Phi(u)&\geq\frac{\|u\|_{\vec{p}(\cdot)}^{P_{-}^{-}}}{P_{+}^{+}N^{P_{-}^{-}-1}}
+\frac{1}{2r^{+}}|u|_{h,r(\cdot)}^{r^{-}}-C_{2}
\geq C_{3}\big(\|u\|_{\vec{p}(\cdot)}+|u|_{h,r(\cdot)}\big)-C_{2}=C_{3}\|u\|_{E}-C_{2},
\end{align*}
where $C_{3}=\min\bigg\{\frac{1}{P_{+}^{+}N^{P_{-}^{-}-1}},\frac{1}{2r^{+}}\bigg\}$. We infer that
$\Phi(u)\rightarrow\infty$ as $\|u\|_{E}\rightarrow\infty$. This means that
$\Phi$ is coercive on $E$.
\qed

\begin{lem}
\label{lemma2}
Assume that $(u_{n})$ is a sequence  in $E$ such that $\Phi(u_{n})$ is bounded. Then there exists a subsequence of $(u_{n})$, still denoted by $(u_{n})$, which converges weakly in $E$ to some $u_{0}\in E$ and
$$\Phi(u_{0})\leq\liminf_{n\rightarrow\infty}\Phi(u_{n}).$$
\end{lem}

{\it Proof.}
Since $\Phi(u_{n})$ is bounded, the inequality \eqref{Phi is bounded} implies that
$\sum_{i=1}^{N}\int_{\Omega}|\partial_{x_{i}}u_{n}|^{p_{i}(x)}dx$ and
$\int_{\Omega}h(x)|u_{n}|^{r(x)}dx$ are bounded.

Let $(u_{n})$ be such that $|\partial_{x_{i}}u_{n}|_{p_{i}(\cdot)}>1$  for all $i\in\{1,...,N\}$, wherefrom
$\|u_{n}\|_{\vec{p}(\cdot)}>1$. By \eqref{|u|>1} we get
$$1<|\partial_{x_{i}}u_{n}|_{p_{i}(\cdot)}<\sum_{i=1}^{N}|\partial_{x_{i}}u_{n}|_{p_{i}(\cdot)}
<\sum_{i=1}^{N}|\partial_{x_{i}}u_{n}|_{p_{i}(\cdot)}^{p_{i}^{-}}
\leq\sum_{i=1}^{N}\int_{\Omega}|\partial_{x_{i}}u_{n}|^{p_{i}(\cdot)}dx,$$
hence $\|u_{n}\|_{\vec{p}(\cdot)}$ is bounded.

Also, let $v_{n}\in L^{r(\cdot)}(\Omega)$, $r\in C_{+}(\overline{\Omega})$, provided that $|v_{n}|_{r(\cdot)}>1$.
Therefore, by inequality \eqref{|u|>1} and taking $v_{n}(x)=h(x)^{\frac{1}{r(x)}}u_{n}(x)$ we obtain
$$1<|u_{n}|_{h,r(\cdot)}<|u_{n}|_{h,r(\cdot)}^{r^{-}}\leq\int_{\Omega}h(x)|u_{n}|^{r(x)}dx,$$
whence $|u_{n}|_{h,r(\cdot)}$ is bounded. Hence $\|u_{n}\|_{E}$ is bounded. Consequently, there exists a
subsequence of $(u_{n})$, relabeled again $(u_{n})$, which converges weakly to some $u_{0}\in E$.
In fact, there exists $u_{0}\in E$ so that
\begin{align*}
&u_{n}\rightharpoonup u_{0} \ \text{in} \ W_{0}^{1,p(\cdot)}(\Omega),\\
&u_{n}\rightarrow u_{0} \ \text{in} \ L_{h}^{r(\cdot)}(\Omega).
\end{align*}
We define
$$F(x,u)=\frac{\lambda}{q(x)}|u|^{q(x)}-\frac{h(x)}{r(x)}|u|^{r(x)}$$
and
$$f(x,u)=F_{u}(x,u)=\lambda|u|^{q(x)-2}u-h(x)|u|^{r(x)-2}u.$$
We notice that
$$f_{u}(x,u)=\lambda(q(x)-1)|u|^{q(x)-2}-h(x)(r(x)-1)|u|^{r(x)-2}.$$
Using again \eqref{k1k2} for $k_{1}=\lambda(q(x)-1)$, $k_{2}=h(x)(r(x)-1)$, $q=q(x)-2$ and $r=r(x)-2$ it follows that
\begin{align*}
f_{u}(x,u)&\leq C\lambda(q(x)-1)\bigg(\frac{\lambda(q(x)-1)}{h(x)(r(x)-1)}\bigg)^{\frac{q(x)-2}{r(x)-q(x)}}\\
&=C\bigg(\frac{q(x)-1}{r(x)-1}\bigg)^{\frac{q(x)-2}{r(x)-q(x)}}\cdot(q(x)-1)\cdot
\frac{\lambda^{\frac{r(x)-2}{r(x)-q(x)}}}{h(x)^{\frac{q(x)-2}{r(x)-q(x)}}}.
\end{align*}
Since $2<q(x)<r(x)<P_{-,\infty}$ we can see that
$\bigg(\frac{q(x)-1}{r(x)-1}\bigg)^{\frac{q(x)-2}{r(x)-q(x)}}\cdot(q(x)-1)$ is a bounded expression.
It follows that
\begin{equation}
\label{f_u}
f_{u}(x,u)\leq C_{1}\frac{\lambda^{\frac{r(x)-2}{r(x)-q(x)}}}{h(x)^{\frac{q(x)-2}{r(x)-q(x)}}}.
\end{equation}
According to the definitions of $\Phi$ and $F$ we obtain the following estimate
\begin{align}
\label{Fi}
\Phi(u_{0})-\Phi(u_{n})&=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{0}|^{p_{i}(x)}}{p_{i}(x)}dx
-\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{n}|^{p_{i}(x)}}{p_{i}(x)}dx
+\int_{\Omega}[F(x,u_{n})-F(x,u_{0})]dx.
\end{align}
Clearly the following
\begin{align*}
\int_{0}^{s}f_{u}(x,u_{0}+t(u_{n}-u_{0}))dt&=\frac{1}{u_{n}-u_{0}}[f(x,u_{0}+s(u_{n}-u_{0}))-f(x,u_{0})]\\
&=\frac{1}{u_{n}-u_{0}}[F_{u}(x,u_{0}+s(u_{n}-u_{0}))-F_{u}(x,u_{0})]
\end{align*}
hold true.
Integrating the previous relation over $[0,1]$ we find that
\begin{align*}
\int_{0}^{1}\int_{0}^{s}f_{u}(x,u_{0}+t(u_{n}-u_{0}))dt\, ds&=
\frac{1}{u_{n}-u_{0}}\int_{0}^{1}[F_{u}(x,u_{0}+s(u_{n}-u_{0}))-F_{u}(x,u_{0})]ds\\
&=\frac{1}{(u_{n}-u_{0})^{2}}[F(x,u_{n})-F(x,u_{0})]-\frac{f(x,u_{0})}{u_{n}-u_{0}}.
\end{align*}
Hence
\begin{equation}
\label{F}
F(x,u_{n})-F(x,u_{0})=(u_{n}-u_{0})^{2}\int_{0}^{1}\int_{0}^{s}f_{u}(x,u_{0}+t(u_{n}-u_{0}))dt\, ds
+(u_{n}-u_{0})f(x,u_{0}).
\end{equation}
Combining the relations \eqref{Fi} and \eqref{F}, then applying \eqref{f_u} we infer that
\begin{align}
\label{Fi'}
\Phi(u_{0})-\Phi(u_{n})=&\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{0}|^{p_{i}(x)}}{p_{i}(x)}dx
-\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{n}|^{p_{i}(x)}}{p_{i}(x)}dx\nonumber\\
&+\int_{\Omega}(u_{n}-u_{0})^{2}\int_{0}^{1}\int_{0}^{s}f_{u}(x,u_{0}+t(u_{n}-u_{0}))dt\, ds\, dx\nonumber\\
&+\int_{\Omega}(u_{n}-u_{0})f(x,u_{0})dx\nonumber\\
\leq&\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{0}|^{p_{i}(x)}}{p_{i}(x)}dx
-\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{n}|^{p_{i}(x)}}{p_{i}(x)}dx\nonumber\\
&+C_{2}\int_{\Omega}(u_{n}-u_{0})^{2}\frac{\lambda^{\frac{r(x)-2}{r(x)-q(x)}}}{h(x)^{\frac{q(x)-2}{r(x)-q(x)}}}dx
+\int_{\Omega}(u_{n}-u_{0})f(x,u_{0})dx,
\end{align}
where $C_{2}>0$ is a constant.
Now, we need to demonstrate that the last two integrals converge to $0$ as $n\rightarrow\infty$.

We define $J:E\rightarrow\RR$ by
$$J(v)=\int_{\Omega}f(x,u_{0})v\, dx.$$
Evidently, $J$ is linear. We want to prove that $J$ is also continuous. We have
\begin{align}
\label{J cont}
|J(v)|&\leq\int_{\Omega}|f(x,u_{0})v|dx=
\int_{\Omega}\big|\lambda|u_{0}|^{q(x)-2}u_{0}-h(x)|u_{0}|^{r(x)-2}u_{0}\big|\cdot|v|dx\nonumber\\
&\leq\lambda\int_{\Omega}|u_{0}|^{q(x)-1}|v|dx+\int_{\Omega}h(x)|u_{0}|^{r(x)-1}|v|dx.
\end{align}
Applying the H\"{o}lder type inequality \eqref{Holder} we find that
$$\int_{\Omega}|u_{0}|^{q(x)-1}|v|dx\leq\Bigg(\frac{1}{\big(\frac{q}{q-1}\big)^{-}}+\frac{1}{q^{-}}\Bigg)
\big||u_{0}|^{q(x)-1}\big|_{\frac{q(\cdot)}{q(\cdot)-1}}|v|_{q(\cdot)}.$$
On the other hand, the embedding $W_{0}^{1,\vec{p}(\cdot)}(\Omega)\hookrightarrow L^{q(\cdot)}(\Omega)$
is continuous and, thus, there exists a positive constant $C_{0}$ such that
$$|v|_{q(\cdot)}\leq C_{0}\|v\|_{\vec{p}(\cdot)}, \ \ \forall v\in W_{0}^{1,\vec{p}(\cdot)}(\Omega).$$
We have also
$$\|v\|_{\vec{p}(\cdot)}\leq\|v\|_{E}.$$
The last three inequalities imply that
\begin{equation}
\label{cont1}
\int_{\Omega}|u_{0}|^{q(x)-1}|v|dx\leq \widetilde{C}\|v\|_{E},
\end{equation}
where $\widetilde{C}>0$ is a constant.
Using again H\"{o}lder's inequality \eqref{Holder} we have
\begin{align}
\label{cont2}
\int_{\Omega}h(x)|u_{0}|^{r(x)-1}|v|dx&=
\int_{\Omega}\big(h(x)^{\frac{r(x)-1}{r(x)}}|u_{0}|^{r(x)-1}\big)\big(h(x)^{\frac{1}{r(x)}}|v|\big)dx\nonumber\\
&\leq\Bigg(\frac{1}{\big(\frac{r}{r-1}\big)^{-}}+\frac{1}{r^{-}}\Bigg)
\bigg|h(x)^{\frac{r(x)-1}{r(x)}}|u_{0}|^{r(x)-1}\bigg|_{\frac{r(\cdot)}{r(\cdot)-1}}
\bigg|h(x)^{\frac{1}{r(x)}}|v|\bigg|_{r(\cdot)}\nonumber\\
&=\overline{C}|v|_{h,r(\cdot)}\leq\overline{C}\|v\|_{E},
\end{align}
where $\overline{C}$ is a positive constant.

So, from \eqref{J cont}, \eqref{cont1} and \eqref{cont2} we can see that there exists a constant $\widehat{C}>0$ so that
$$|J(v)|\leq\widehat{C}\|v\|_{E}, \ \ \forall v\in E,$$
that is $J$ is continuous. Now, considering that $u_{n}\rightharpoonup u_{0}$ in $E$ and $J$ is linear and continuous we conclude that
$$J(u_{n})\rightarrow J(u_{0}).$$
This shows that
\begin{equation}
\label{lim}
\lim_{n\rightarrow\infty}\int_{\Omega}f(x,u_{0})(u_{n}-u_{0})dx=0.
\end{equation}
Next, we prove that
\begin{equation}
\label{(u_n-u_0)^2}
\lim_{n\rightarrow\infty}
\int_{\Omega}(u_{n}-u_{0})^{2}\frac{\lambda^{\frac{r(x)-2}{r(x)-q(x)}}}{h(x)^{\frac{q(x)-2}{r(x)-q(x)}}}dx=0.
\end{equation}
We know that $u_{n}\rightharpoonup u_{0}$ in $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$. Since the embedding $W_{0}^{1,\vec{p}(\cdot)}(\Omega)\hookrightarrow L^{r(\cdot)}(\Omega)$ is compact, it follows that $u_{n}\rightarrow u_{0}$ in $L^{r(\cdot)}(\Omega)$, that is,
$$\int_{\Omega}|u_{n}-u_{0}|^{r(x)}dx\rightarrow0.$$
Therefore
$$\int_{\Omega}\big(|u_{n}-u_{0}|^{2}\big)^{\frac{r(x)}{2}}dx<\infty$$
or equivalently
$$(u_{n}-u_{0})^{2}\in L^{\frac{r(\cdot)}{2}}(\Omega).$$
We notice that the condition \eqref{h2} may be equivalently
reformulated as $\frac{\lambda^{\frac{r(x)-2}{r(x)-q(x)}}}{h(x)^{\frac{q(x)-2}{r(x)-q(x)}}}\in L^{\frac{r(\cdot)}{r(\cdot)-2}}(\Omega)$.
Hence, we get
$$\int_{\Omega}(u_{n}-u_{0})^{2}\frac{\lambda^{\frac{r(x)-2}{r(x)-q(x)}}}{h(x)^{\frac{q(x)-2}{r(x)-q(x)}}}dx
\leq\Bigg(\frac{1}{\big(\frac{r}{2}\big)^{-}}+\frac{1}{\big(\frac{r}{r-2}\big)^{-}}\Bigg)
\Bigg|\frac{\lambda^{\frac{r(x)-2}{r(x)-q(x)}}}{h(x)^{\frac{q(x)-2}{r(x)-q(x)}}}\Bigg|_{\frac{r(\cdot)}{r(\cdot)-2}} \big|\big(u_{n}-u_{0}\big)^{2}\big|_{\frac{r(\cdot)}{2}}$$
via H\"{o}lder type inequality \eqref{Holder}.
We notice that also
$$\rho_{\frac{r(\cdot)}{2}}((u_{n}-u_{0})^{2})=\int_{\Omega}|(u_{n}-u_{0})^{2}|^{\frac{r(x)}{2}}dx=
\int_{\Omega}|u_{n}-u_{0}|^{r(x)}dx\rightarrow0.$$
The above relation together with \eqref{conv 0} implies
$$\big|\big(u_{n}-u_{0}\big)^{2}\big|_{\frac{r(\cdot)}{2}}\rightarrow0.$$
Therefore, it turns out that \eqref{(u_n-u_0)^2} holds true.

Now, to conclude Lemma \ref{lemma2} we prove that the functional
$\Lambda:W_{0}^{1,\vec{p}(\cdot)}(\Omega)\rightarrow\RR$,
$$\Lambda(u)=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx$$
is convex. Indeed, taking in consideration that the function
$$[0,\infty)\ni t\rightarrow t^{\gamma}$$
is convex for each $\gamma>1$ it follows that for any $x\in\Omega$ fixed it the inequality
\begin{equation}
\label{alpha+beta}
\bigg|\frac{\alpha+\beta}{2}\bigg|^{p_{i}(x)}\leq\bigg|\frac{|\alpha|+|\beta|}{2}\bigg|^{p_{i}(x)}\leq\frac{1}{2}|\alpha|^{p_{i}(x)}
+\frac{1}{2}|\beta|^{p_{i}(x)}, \ \forall \ \alpha,\beta\in\RR, \ i\in\{1,...,N\}
\end{equation}
holds. Using \eqref{alpha+beta} we obtain
\begin{equation*}
\bigg|\frac{\partial_{x_{i}}u+\partial_{x_{i}}v}{2}\bigg|^{p_{i}(x)}\leq\frac{1}{2}|\partial_{x_{i}}u|^{p_{i}(x)}+
\frac{1}{2}|\partial_{x_{i}}v|^{p_{i}(x)}, \ \forall \ u,v\in W_{0}^{1,\vec{p}(\cdot)}(\Omega), \ x\in\Omega, \ i\in\{1,...,N\}.
\end{equation*}
Multiplying by $\frac{1}{p_{i}(x)}$, summing from $1$ to $N$ and integrating over $\Omega$ we deduce that
\begin{equation*}
\Lambda\bigg(\frac{u+v}{2}\bigg)\leq\frac{1}{2}\Lambda(u)+\frac{1}{2}\Lambda(v), \ \forall \ u,v\in W_{0}^{1,\vec{p}(\cdot)}(\Omega).
\end{equation*}
Therefore $\Lambda$ is convex on $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$. A way to prove that the functional $\Lambda$ is
weakly lower semicontinuous  on $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ is to show that this is lower semicontinuous on
$W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ via Corollary III.8 in \cite{bib17}. To this aim, we fix $u\in W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ and $\varepsilon>0$. Let $v\in W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ be arbitrary.
The fact that $\Lambda$ is convex and inequality \eqref{Holder} holds lead us to
\begin{align*}
\Lambda(v)&\geq\Lambda(u)+\langle\Lambda'(u),v-u\rangle\\
&=\Lambda(u)+\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)-2}\partial_{x_{i}}u\partial_{x_{i}}(v-u)dx\\
&\geq\Lambda(u)-\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)-1}|\partial_{x_{i}}(v-u)|dx\\
&\geq\Lambda(u)-c_{1}\sum_{i=1}^{N}\big||\partial_{x_{i}}u|^{p_{i}(x)-1}\big|_{\frac{p_{i}(\cdot)}{p_{i}(\cdot)-1}}
|\partial_{x_{i}}(v-u)|_{p_{i}(\cdot)}\\
&\geq\Lambda(u)-c_{2}\|v-u\|_{\vec{p}(\cdot)}\\
&\geq\Lambda(u)-\varepsilon
\end{align*}
for any $v\in W_{0}^{1,\vec{p}(\cdot)}$ with $\|v-u\|_{\vec{p}(\cdot)}<\varepsilon/c_{2}$, where $c_{1}, c_{2}$
are two positive constants. Thus $\Lambda$ is lower semicontinuous on $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ and so weakly lower semicontinuous.
This means that
$$\liminf_{n\rightarrow\infty}\Lambda(u_{n})\geq\Lambda(u_{0}),$$
that is
$$\liminf_{n\rightarrow\infty}\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{n}|^{p_{i}(x)}}{p_{i}(x)}dx
\geq\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{0}|^{p_{i}(x)}}{p_{i}(x)}dx.$$
Passing to the limit in \eqref{Fi'} and using \eqref{lim}, \eqref{(u_n-u_0)^2} and the above inequality it follows that
$$\liminf_{n\rightarrow\infty}\Phi(u_{n})\geq\Phi(u_{0}).$$
We conclude that Lemma \ref{lemma2} holds true.
\qed

\textsc{Proof of Theorem \ref{thm2}.} By Lemmas \ref{lemma1}, \ref{lemma2} and Theorem 1.2 in \cite{bib6} we deduce that there exists a global minimizer  of $\Phi$, that is
$$\Phi(u)=\inf_{v\in E}\Phi(v).$$
It is obvious that $u$ is weak solution of problem \eqref{pb}. Our goal is to show that $u\not\equiv0$ in $E$. For this purpose, we will prove that $\inf_{E}\Phi<0$ providing that $\lambda$ is large enough.

Set
$$\overline{\lambda}=\inf\bigg\{q^{+}\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx
+q^{+}\int_{\Omega}\frac{h(x)}{r(x)}|u|^{r(x)}dx; \ u\in E, \ \int_{\Omega}|u|^{q(x)}dx=1\bigg\}.$$

We want to point out that $\overline{\lambda}>0$. For all $u\in E$ with $\int_{\Omega}|u|^{q(x)}dx=1$, H\"{o}lder type inequality \eqref{Holder} yields
\begin{equation}
\label{lambda}
\lambda=\int_{\Omega}\frac{\lambda}{h(x)^{\frac{q(x)}{r(x)}}}h(x)^{\frac{q(x)}{r(x)}}|u|^{q(x)}dx\leq
\bigg|\frac{\lambda}{h(x)^{{\frac{q(x)}{r(x)}}}}\bigg|_{\frac{r(\cdot)}{r(\cdot)-q(\cdot)}}\bigg|
h(x)^{\frac{q(x)}{r(x)}}|u|^{q(x)}\bigg|_{\frac{r(\cdot)}{q(\cdot)}}.
\end{equation}
Next, we focus our attention on the case when $u\in E$ such that
$$\bigg|h(x)^{\frac{q(x)}{r(x)}}|u|^{q(x)}\bigg|_{\frac{r(\cdot)}{q(\cdot)}}>1.$$
Thus, by \eqref{|u|>1}, \eqref{|u|<1} and \eqref{lambda} it follows that
\begin{equation*}
\lambda\leq\Bigg(\int_{\Omega}\frac{\lambda^{\frac{r(x)}{r(x)-q(x)}}}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}dx\Bigg)
^{1/(\frac{r}{r-q})^{\pm}}\bigg(\int_{\Omega}h(x)|u|^{r(x)}dx\bigg)^
{1/(\frac{r}{q})^{-}}.
\end{equation*}
Hence, we get
\begin{equation*}
\int_{\Omega}h(x)|u|^{r(x)}dx\geq\frac{\lambda^{(\frac{r}{q})^{-}}}
{\Bigg(\displaystyle\int_{\Omega}\frac{\lambda^{\frac{r(x)}{r(x)-q(x)}}}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}dx\Bigg)^
{(\frac{r}{q})^{-}/(\frac{r}{r-q})^{\pm}}}.
\end{equation*}
Therefore
$$\overline{\lambda}\geq q^{+}\int_{\Omega}\frac{h(x)}{r(x)}|u|^{r(x)}dx\geq\frac{q^{+}}{r^{+}}\lambda^{(\frac{r}{q})^{-}}\Bigg(\int_{\Omega}
\frac{\lambda^{\frac{r(x)}{r(x)-q(x)}}}{h(x)^{\frac{q(x)}{r(x)-q(x)}}}dx\Bigg)
^{-{(\frac{r}{q})^{-}/(\frac{r}{r-q})^{\pm}}}>0.$$
Let $\lambda>\overline{\lambda}$. Then there exists a function $u_{1}\in E$ with $\int_{\Omega}|u_{1}|^{q(x)}dx=1$ so that
$$\lambda\int_{\Omega}|u_{1}|^{q(x)}dx=\lambda>q^{+}\int_{\Omega}\sum_{i=1}^{N}
\frac{|\partial_{x_{i}}u_{1}|^{p_{i}(x)}}{p_{i}(x)}dx+q^{+}\int_{\Omega}\frac{h(x)}{r(x)}|u_{1}|^{r(x)}dx.$$
Thus, we deduce that
$$\lambda\int_{\Omega}\frac{1}{q(x)}|u_{1}|^{q(x)}dx\geq\frac{\lambda}{q^{+}}\int_{\Omega}|u_{1}|^{q(x)}dx>
\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{1}|^{p_{i}(x)}}{p_{i}(x)}dx+\int_{\Omega}\frac{h(x)}{r(x)}|u_{1}|^{r(x)}dx.$$
Finally, we obtain
$$\Phi(u_{1})=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u_{1}|^{p_{i}(x)}}{p_{i}(x)}dx-
\lambda\int_{\Omega}\frac{1}{q(x)}|u_{1}|^{q(x)}dx+\int_{\Omega}\frac{h(x)}{r(x)}|u_{1}|^{r(x)}dx<0,$$
so we infer that $\inf_{u\in E}\Phi(u)<0$. In conclusion, there exists $\lambda_{0}=\overline{\lambda}$ so that the problem \eqref{pb} has a nontrivial weak solution, namely $u_{1}\in E$, for every $\lambda>\lambda_{0}$, satisfying $\Phi(u_{1})<0$. Since $\Phi(u_{1})=\Phi(|u_{1}|)$ we may assume that $u_{1}\geq0$ almost everywhere in $\Omega$.
\qed

Next, we will try to find a second nontrivial weak solution for problem \eqref{pb}. To this aim, we fix $\lambda\geq\lambda_{0}$ and set
\[ g(x,t) = \left\{ \begin{array}{l l} 0, & \quad \text{for} \ t<0\\
\lambda t^{q(x)-1}-h(x)t^{r(x)-1}, & \quad \text{for} \ 0\leq t\leq u_{1}(x)\\
\lambda u_{1}(x)^{q(x)-1}-h(x)u_{1}(x)^{r(x)-1}, & \quad \text{for} \ t>u_{1}(x)
\end{array} \right. \]
and
$$G(x,t)=\int_{0}^{t}g(x,s)ds.$$
We define the functional $\Psi:E\rightarrow\RR$ by
$$\Psi(u)=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx
-\int_{\Omega}G(x,u)dx.$$
By standard arguments, $\Psi\in C^{1}(E,\RR)$ and the derivative is given by
$$\langle\Psi'(u),v\rangle=\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)-2}
\partial_{x_{i}}u\partial_{x_{i}}v\, dx-\int_{\Omega}g(x,u)v\, dx$$
for all $u,v\in E$. It is obvious that if $u$ is a critical point of $\Psi$ then $u\geq0$ almost everywhere in $\Omega$.

Let us prove now the following lemma.
\begin{lem}
If $u$ is a critical point of $\Psi$ then $u\leq u_{1}$.
\end{lem}

{\it Proof.}
Denote by $v^{+}(x)=\max\{v(x),0\}$ the positive part of $v$. Using Theorem 7.6 in \cite{bib7} it follows that if $v\in E$ then $v^{+}\in E$. Next, we will apply the following inequality (see formula (2.2) in \cite{bib19})
$$(|\xi_{i}|^{s_{i}-2}\xi_{i}-|\psi_{i}|^{s_{i}-2}\psi_{i})(\xi_{i}-\psi_{i})
\geq2^{-s_{i}}|\xi_{i}-\psi_{i}|^{s_{i}}, \ \forall \ \xi_{i},\psi_{i}\in \RR$$
valid for each $s_{i}\geq2$. We have
\begin{align*}
0&=\langle\Psi'(u),(u-u_{1})^{+}\rangle-\langle\Phi'(u_{1}),(u-u_{1})^{+}\rangle\\
&=\int_{\Omega}\sum_{i=1}^{N}\big(|\partial_{x_{i}}u|^{p_{i}(x)-2}\partial_{x_{i}}u-
|\partial_{x_{i}}u_{1}|^{p_{i}(x)-2}\partial_{x_{i}}u_{1}\big)\partial_{x_{i}}(u-u_{1})^{+}dx\\
&\quad{}-\int_{\Omega}\big[g(x,u)-\lambda u_{1}^{q(x)-1}+h(x)u_{1}^{r(x)-1}\big](u-u_{1})^{+}dx\\
&=\int_{[u>u_{1}]}\sum_{i=1}^{N}\big(|\partial_{x_{i}}u|^{p_{i}(x)-2}\partial_{x_{i}}u-
|\partial_{x_{i}}u_{1}|^{p_{i}(x)-2}\partial_{x_{i}}u_{1}\big)(\partial_{x_{i}}u-\partial_{x_{i}}u_{1})dx\\
&\geq\int_{[u>u_{1}]}\sum_{i=1}^{N}2^{-p_{i}(x)}|\partial_{x_{i}}u-\partial_{x_{i}}u_{1}|^{p_{i}(x)}dx\geq0.
\end{align*}
This holds if and only if $(\partial_{x_{i}}u-\partial_{x_{i}}u_{1})(x)=0$ for all $x\in\omega:=\{y\in\Omega; \ u(y)>u_{1}(y)\}$, \ $i\in\{1,...,N\}$. Therefore $|\partial_{x_{i}}u-\partial_{x_{i}}u_{1}|_{p_{i}(\cdot)}<1$ in $\omega$. Thus, by \eqref{|u|<1}
$$|\partial_{x_{i}}u-\partial_{x_{i}}u_{1}|_{p_{i}(\cdot)}^{p_{i}^{+}}
\leq\int_{\omega}|\partial_{x_{i}}u-\partial_{x_{i}}u_{1}|^{p_{i}(x)}dx
\leq|\partial_{x_{i}}u-\partial_{x_{i}}u_{1}|_{p_{i}(\cdot)}^{p_{i}^{-}}.$$
Since $\partial_{x_{i}}u-\partial_{x_{i}}u_{1}=0$ in $\omega$ we obtain that
$|\partial_{x_{i}}u-\partial_{x_{i}}u_{1}|_{p_{i}(\cdot)}\leq0$ and this is a contradiction with
$|\partial_{x_{i}}u-\partial_{x_{i}}u_{1}|_{p_{i}(\cdot)}>0$. We conclude that $u\leq u_{1}$ in $\Omega$. This completes the proof.
\qed

In what follows our goal is to determine a critical point $u_{2}\in E$ of $\Psi$ such that $\Psi(u_{2})>0$ via mountain pass theorem. By the previous lemma we deduce that $0\leq u_{2}\leq u_{1}$ in $\Omega$. Hence
$$g(x,u_{2})=\lambda u_{2}^{q(x)-1}-h(x)u_{2}^{r(x)-1} \ \text{ and }\ G(x,u_{2})=\frac{\lambda}{q(x)}u_{2}^{q(x)}-\frac{h(x)}{r(x)}u_{2}^{r(x)}$$
and we find that
$$\Psi(u_{2})=\Phi(u_{2}) \ \text{ and } \ \Psi'(u_{2})=\Phi'(u_{2}).$$
So, we have the following
$$\Phi(u_{2})=\Psi(u_{2})>0=\Phi(0)>\Phi(u_{1})$$
and
$$0=\Psi'(u_{2})=\Phi'(u_{2}),$$
that is $u_{2}$ is a weak solution of problem \eqref{pb} such that $0\leq u_{2}\leq u_{1}$, $u_{2}\neq0$ and $u_{2}\neq u_{1}$.

Now, we focus our attention to get $u_{2}$ described above.
\begin{lem}
\label{lem cu Psi>a}
There exists $\rho\in(0,\|u_{1}\|)$ and $a>0$ so that $\Psi(u)\geq a$, for every $u\in E$ with $\|u\|=\rho$.
\end{lem}

{\it Proof.}
For every $u\in E$ we have
\begin{align}
\label{Psi}
\Psi(u)&=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx-\int_{[u>u_{1}]}G(x,u)dx-\int_{[u\leq u_{1}]}G(x,u)dx\nonumber\\
&=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx-\lambda\int_{[u>u_{1}]} u_{1}^{q(x)-1}u\, dx+\int_{[u>u_{1}]}h(x)u_{1}^{r(x)-1}u\, dx\nonumber\\
&\quad-\lambda\int_{[u\leq u_{1}]}\frac{1}{q(x)}u^{q(x)}\, dx+\int_{[u\leq u_{1}]}\frac{h(x)}{r(x)}u^{r(x)}dx\nonumber\\
&>\frac{1}{P_{+}^{+}}\sum_{i=1}^{N}\int_{\Omega}|\partial_{x_{i}}u|^{p_{i}(x)}dx
-\lambda\int_{[u>u_{1}]}u^{q(x)}dx-\frac{\lambda}{q^{-}}\int_{[u\leq u_{1}]}u^{q(x)}dx\nonumber\\
&>\frac{1}{P_{+}^{+}}\sum_{i=1}^{N}\int_{\Omega}|\partial_{x_{i}}u|^{p_{i}(x)}dx-\lambda\int_{\Omega}|u|^{q(x)}dx.
\end{align}
We know that $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ is continuously embedded in $L^{q(\cdot)}(\Omega)$, thus there exists a constant $L>1$ so that
\begin{equation}
\label{cont3}
|u|_{q(\cdot)}\leq L\cdot\|u\|_{\vec{p}(\cdot)}, \ \ \forall u\in W_{0}^{1,\vec{p}(\cdot)}(\Omega).
\end{equation}
We fix $\rho\in(0,1)$ such that $\rho<\frac{1}{L}$. Then the previous relation implies
$$|u|_{q(\cdot)}<1 \ \text{for all} \ u\in W_{0}^{1,\vec{p}(\cdot)}(\Omega) \ \text{with} \ \|u\|_{\vec{p}(\cdot)}=\rho.$$
Evidently, $|\partial_{x_{i}}u|_{p_{i}(\cdot)}<1$. In this case we have shown already in the proof of Theorem \ref{thm1} that
\begin{equation*}
\sum_{i=1}^{N}\int_{\Omega}|\partial_{x_{i}}u|^{p_{i}(x)}dx
\geq\frac{\|u\|_{\vec{p}(\cdot)}^{P_{+}^{+}}}{N^{P_{+}^{+}-1}}.
\end{equation*}
On the other hand, \eqref{|u|<1} and \eqref{cont3} imply
$$\lambda\int_{\Omega}|u|^{q(x)}dx\leq L_{1}\|u\|_{\vec{p}(\cdot)}^{q^{-}},$$
where $L_{1}=\lambda L^{q^{-}}$. So, considering \eqref{Psi} and the last two inequalities we have
$$\Psi(u)>\frac{\|u\|_{\vec{p}(\cdot)}^{P_{+}^{+}}}{P_{+}^{+}N^{P_{+}^{+}-1}}
-L_{1}\|u\|_{\vec{p}(\cdot)}^{q^{-}}
=\|u\|_{\vec{p}(\cdot)}^{P_{+}^{+}}\Bigg(
\frac{1}{P_{+}^{+}N^{P_{+}^{+}-1}}-L_{1}\|u\|_{\vec{p}(\cdot)}^{q^{-}-P_{+}^{+}}\Bigg),$$
for all $u\in E$ with $\|u\|_{\vec{p}(\cdot)}=\rho$.

We notice that the function $g:[0,1]\rightarrow\RR$ defined by
$$g(t)=\frac{1}{P_{+}^{+}N^{P_{+}^{+}-1}}-L_{1}t^{q^{-}-P_{+}^{+}}$$
is positive in a neighborhood of the origin, so that the choice of $\rho\in(0,1)$ is
so small that $a=\rho^{P_{+}^{+}}g(\rho)>0$. We conclude that lemma holds true.
\qed

\begin{lem}
\label{coer}
The functional $\Psi$ is coercive in $E$.
\end{lem}

{\it Proof.}
For every $u \in E$ we have
\begin{align}
\Psi(u)&=\int_{\Omega}\sum_{i=1}^{N}\frac{|\partial_{x_{i}}u|^{p_{i}(x)}}{p_{i}(x)}dx-\lambda\int_{[u>u_{1}]} u_{1}^{q(x)-1}u\, dx+\int_{[u>u_{1}]}h(x)u_{1}^{r(x)-1}u\, dx\nonumber\\
&\quad-\lambda\int_{[u\leq u_{1}]}\frac{1}{q(x)}u^{q(x)}\, dx+\int_{[u\leq u_{1}]}\frac{h(x)}{r(x)}u^{r(x)}dx\nonumber\\
&>\frac{1}{P_{+}^{+}}\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx-\lambda\int_{[u>u_{1}]}u_{1}^{q(x)}dx-\frac{\lambda}{q^{-}}\int_{[u\leq u_{1}]}u_{1}^{q(x)}dx\nonumber\\
&>\frac{1}{P_{+}^{+}}\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx-\lambda\int_{\Omega}u_{1}^{q(x)}dx\nonumber\\
&=\frac{1}{P_{+}^{+}}\int_{\Omega}\sum_{i=1}^{N}|\partial_{x_{i}}u|^{p_{i}(x)}dx-L_{2}\nonumber,
\end{align}
where $L_{2}>0$ is a constant. Taking $u\in E$ such that $\|u\|_{\vec{p}(\cdot)}>1$ we obtain
$$\Psi(u)>\frac{\|u\|_{\vec{p}(\cdot)}^{P_{-}^{-}}}{P_{+}^{+}N^{P_{-}^{-}-1}}-L_{2}.$$
We conclude that $\Psi(u)\rightarrow\infty$ as $\|u\|_{\vec{p}(\cdot)}\rightarrow\infty$. In other words, $\Psi$ is coercive in $E$, completing the proof.
\qed

\textsc{Proof of Theorem \ref{thm2} completed.}
By Lemma \ref{lem cu Psi>a} and the mountain pass theorem (see \cite{bib8} with the variant given by Theorem 1.15 in \cite{bib9}) we derive that there exists a sequence $(u_{n})\subset E$ so that
\begin{equation}
\label{mountain}
\Psi(u_{n})\rightarrow c>0 \ \text{ and } \ \Psi'(u_{n})\rightarrow0
\end{equation}
where
$$c=\inf_{\gamma\in\Gamma}\max_{t\in[0,1]}\Psi(\gamma(t))$$
and
$$\Gamma=\{\gamma\in C([0,1],E); \ \gamma(0)=0,\ \gamma(1)=u_{1}\}.$$
Keeping in mind \eqref{mountain} and relying on Lemma \ref{coer} we infer that $(u_{n})$ is bounded and, thus, passing eventually to a subsequence, relabeled again $(u_{n})$, we may assume that there exists $u_{2}\in E$ so that $u_{n}\rightharpoonup u_{2}$. Standard arguments, based on the Sobolev embeddings, help us to see that
$$\lim_{n\rightarrow\infty}\langle\Psi'(u_{n}),v\rangle=\langle\Psi'(u_{2}),v\rangle,$$
for every $v\in C_{0}^{\infty}(\Omega)$. The above piece of information together with the fact that $E\subset W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ and $C_{0}^{\infty}(\Omega)$ is dense in $W_{0}^{1,\vec{p}(\cdot)}(\Omega)$ yield to the fact that $u_{2}$ is a weak solution of problem \eqref{pb}. Thus, our conclusion is that problem \eqref{pb} has at least two nontrivial weak solutions and the proof of Theorem \ref{thm2} is now complete.
\qed

\medskip
{\bf Acknowledgments.} The authors are grateful to the anonymous referee for a detailed and insightful report on an earlier version of this paper and for valuable comments and suggestions.

% Bibliography
%-----------------------------------------------------------------


\end{document} 